Author's response to reviews

Title: DanQi Pill protects against heart failure through the arachidonic acid metabolism pathway by balancing different cyclooxygenase patterns

Authors:

Yong Wang (doctor_wangyong@163.com)
Chun Li (34451574@qq.com)
Yang Zhong Liu (qiufengqihuangmo@sina.com)
Jiao Tian Shi (tianjiaoshi1988@sina.cn)
Yan Qi Wang (wangqybcem@163.com)
Dong Li (lidongbprc@163.com)
Yan Wu (yanwubucm@163.com)
Jing Han (hanjingbucm@sina.com)
Zhen Shu Guo (shuzhenguo@sina.cn)
Hua Bing Tang (binghuatang@sina.cn)
Wei Wang (wangwei_8@sina.cn)

Version: 2 Date: 24 September 2013

Author's response to reviews:

We revised our manuscript as you suggested as follows:
1. We added the Competing Interests sections in the revised manuscript;
2. We formatted our Abstract into subsections in accordance with formatting guidelines of BMC Complementary and Alternative Medicine.

Thanks a lot for your work on our paper. If there is any questions, please contact me at any time.

Best Regards

Wang Wei

Beijing University of Chinese Medicine